Author's response to reviews

Title: A randomised controlled trial of an intervention to increase the implementation of a healthy canteen policy in Australian primary schools: study protocol.

Authors:

Luke Wolfenden Dr (luke.wolfenden@hnehealth.nsw.gov.au)
Nicole K Nathan (nicole.nathan@hnehealth.nsw.gov.au)
Christopher M Williams Dr (christopher.m.williams@hnehealth.nsw.gov.au)
Tessa Delaney (tessa.delaney@hnehealth.nsw.gov.au)
Kathryn L Reilly (kathryn.reilly@hnehealth.nsw.gov.au)
Megan Freund Dr (megan.freund@hnehealth.nsw.gov.au)
Karen Gillham Dr (karen.gillham@hnehealth.nsw.gov.au)
Rachel Sutherland Dr (rachel.sutherland@hnehealth.nsw.gov.au)
Andrew C Bell Dr (colin.bell@deakin.edu.au)
Libby Cambell (libby.campbell@hnehealth.nsw.gov.au)
Serene Yoong Dr (serene.yoong@hnehealth.nsw.gov.au)
Rebecca Wyse (rebecca.wyse@hnehealth.nsw.gov.au)
Lisa M Janssen (lisa.janssen@hnehealth.nsw.gov.au)
Sarah Preecee (sarah.preecee@hnehealth.nsw.gov.au)
Melanie Asmar (melanie.asmar@hnehealth.nsw.gov.au)
John Wiggers (john.wiggers@hnehealth.nsw.gov.au)

Version: 3 Date: 16 September 2014

Author's response to reviews:

There are no revisions to the cover letter.

Minor manuscript alterations were made in the abstract as a result of completing the CONSORT checklist document required by Implementation Science for publication.